Author's response to reviews

Title: The reduction of disability in community-dwelling frail older people: design of a two-arm cluster randomized controlled trial

Authors:

Silke F Metzelthin (s.metzelthin@zw.unimaas.nl)
Erik van Rossum (e.van.rossum@hszuyd.nl)
Luc P de Witte (l.de.witte@hszuyd.nl)
Marike RC Hendriks (marike.hendriks@beoz.unimaas.nl)
Gertrudis IJM Kempen (g.kempen@zw.unimaas.nl)

Version: 6 Date: 23 August 2010

Author's response to reviews: see over
Dear Dr. Norton,

We have made the following final changes:

1. Authors' contributions: We have changed the list of authors in the manuscript according to authors list in the submission system (First name Initial Last name).
2. Table 2: We have removed bullet points.
3. References: reference 29, Submitted. We refer to the article in the main text in the following way: (Authors: Title of manuscript, submitted). We have removed the reference from the reference list.

We hope that all the changes we have made fulfill your expectations.

On behalf of all authors,

Yours sincerely,

Silke Metzelthin, MPhil.